Author’s response to reviews

Title: A rare case of term viable secondary abdominal pregnancy following rupture of rudimentary horn: a case report.

Authors:

Bhandary Amritha (amrithabhandary@rediffmail.com)
Thirunavukkarasu Sumangali (drsumangali_vijay@yahoo.co.in)
Ballal Priya (priyab@yahoo.co.in)
Shedde Deepak (deepaks@yahoo.com)
Rai Sharadha (sharadar@rediffmail.com)

Version: 3 Date: 24 February 2008

Author’s response to reviews:

To,
The Editor,
Journal of medical case reports.

Sir,
I am really sorry for the difference in the age mentioned in the abstract and manuscript. It was a typing mistake. Kindly pardon me for this mistake and the trouble created. I have changed the age of the patient in the main manuscript and resubmitted it again.